Author’s response to reviews

Title: Spatio-temporal analysis of the relationship between meteorological factors and hand footmouth disease in Beijing, China

Authors:

Lin Tian (tianlin@bjmu.edu.cn)
Fengchao Liang (liangfengchao126@126.com)
Meimei Xu (xumeimei2005@163.com)
Lei Jia (lailajia@126.com)
Xiaochuan Pan (xcpan@bjmu.edu.cn)
Archie Clements (director.rsph@anu.edu.au)

Version: 3 Date: 22 Mar 2018

Author’s response to reviews:

Dear editors,

We appreciated the great help and efforts which the editors and the reviewers have given to our manuscript. The editors and reviewers’ suggestions were incorporated into the revised manuscript. We believe we have adequately addressed all the points. We declare that all the maps depicted in our figures were from our work.

All authors declare that they have no potential competing interests regarding the revised submitted manuscript. They also have read the revised manuscript and agree the work is ready for submission to the BMC Infectious Diseases, and accept responsibility for the manuscript’s contents.
Thank you very much for your consideration!

Sincerely yours,

Xiaochuan Pan,  Professor
Department of Occupational and Environmental Health
Peking University School of Public Health
38 Xueyuan Road, Beijing 100191, P.R. China
Email address: xcpan@bjmu.edu.cn